Author's response to reviews

Title: Pediatric Ramsay Hunt syndrome - full recovery with inadvertent use of a lower dose of acyclovir for shorter than usual: a case report

Authors:

Gwinyai Masukume (parturitions@gmail.com)
Sheenah Chibwowa (drsheenah@hotmail.com)
Mbongeni Ndlovu (mbongenigm@yahoo.co.uk)

Version: 4 Date: 8 January 2011

Author's response to reviews: see over
Dear Editors,

We have amended a sentence in the paper as follows – we are of the opinion that this revision is clearer:

“The facial nerve paralysis was grade IV (moderately severe dysfunction), using the House-Brackmann facial nerve grading system (ranging from I to VI, with I indicating normal function and VI indicating total paralysis).”

Before:

‘The facial nerve paralysis was grade IV (moderately severe dysfunction), using the House-Brackmann facial nerve grading system.’

Sincerely yours,

GM, SC & MN